

Frequency, persistence and relation of disease symptoms, psychosomatic comorbidity and daily life impairment after COVID-19: a cohort study in general practice



Alexander Hapfelmeier^{1*}, Jan Donhauser¹, Clara Teusen¹, Stefanie Eck¹, and Antonius Schneider¹

Abstract

Background Long-lasting symptoms with a possible relation to psychosomatic comorbidity have been described following COVID-19. However, data is sparse in general practice. The trial's objective was to investigate the time-dependent frequency of disease symptoms and relation to psychosomatic comorbidity and daily life impairment (DLI).

Methods Comparative cohort study of patients reporting a previous SARS-CoV-2 infection and uninfected controls in general practice. Participants were recruited in 14 general practices in the greater Munich area. Data collection was questionnaire based with a 12 months follow-up. Descriptive statistics, multivariable regression and bivariate correlations were used for analysis.

Results A total of n = 204 cases infected up to 42 months ago (n = 141 Omicron, n = 63 earlier variants), and n = 119 controls were included. Disease symptoms were substantially more prevalent in cases (55–79% vs. 43% within one year of infection). This difference also appeared in the multivariable analysis adjusting for socio-demographics and psychosomatic comorbidity with odds ratios (OR) of 4.15 (p < 0.001) and 3.51 (p = 0.054) for the cohorts with Omicron or earlier variants infection (vs. controls), respectively. It was persistent with earlier variants (OR 1.00 per month, p = 0.903), while a decreasing trend was observed for Omicron (OR 0.89 per month, p < 0.001). DLI was especially correlated with fatigue (r = 0.628).

Conclusion DLI, psychosomatic comorbidity and independently increased disease symptoms require holistic treatment of the patient in general practice according to the bio-psycho-social model. A key role in restoring the daily life capability may be attributed to the symptom fatigue.

Keywords SARS-CoV-2, General practice, Primary health care, Medically unexplained symptoms, Somatoform disorders

*Correspondence: Alexander Hapfelmeier Alexander.Hapfelmeier@tum.de ¹Institute of General Practice and Health Services Research, TUM School of Medicine and Health, Technical University of Munich, Munich, Germany



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicate of the original autory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Deciration waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

Background

In the COVID-19 pandemic, the perspective has changed from an acute emergency situation to the long-term management of a better known and less severe disease, requiring better understanding and treatment of patients with long-term health problems [1, 2]. In this context, fatigue, dyspnoea, difficulty concentrating and many other long-lasting symptoms have been described, usually accompanied by daily life impairment (DLI) and reduced perceived quality of life [3]. Accordingly, long COVID syndrome (LCS) is defined by symptoms persisting over four weeks, while Post-COVID syndrome (PCS) describes new or persisting symptoms over three months after infection, impacting daily functioning [3].

Data on the prevalence of long-lasting symptoms are heterogeneous. Among the many reasons for such deviations are different recruitment of study populations and temporal effects [3, 4]. To name a few examples from the vast body of literature, one population-based study described persistent symptoms in 13.3% of SARS-CoV-2-infected patients 28 days after infection using health insurance registry data [5]. A cohort study reported a 4% prevalence twelve weeks after infection in school children [6]. In a meta-analysis of COVID symptoms including studies from diverse settings and populations [7] and in another cohort study of participants recruited through social media campaigns or personal invitation [8], 15% and 85% of long COVID cases remained affected at twelve months, respectively.

The nature of LCS or PCS has also been widely debated, with some studies suggesting functional responses to the pandemic rather than due to the disease itself [9, 10]. Others have shown that psychosomatic comorbidity is widely associated with long COVID [3, 11, 12]. Risk factors for the disease such as female sex, lack of vaccination, older age, obesity, infection with pre-Omicron variants or various pre-existing medical conditions are also discussed [1, 13–15]. However, the impact of psychosomatic comorbidity and patient selection on primary care remained unclear up to now.

The present study was conducted to investigate the prevalence and persistence of physical symptoms and DLI after SARS-CoV-2 infection with regard to possible risk factors and a possible dependence on psychosomatic comorbidity in general practice. See supplementary data S1 for a plain language summary.

Methods

Study design

This is a prospective comparative cohort study of cases reporting a previous SARS-CoV-2 infection and uninfected controls with a 12 months patient individual follow-up. The study was conducted in general practice. Data collection was questionnaire based. Each patient filled out a questionnaire upon enrolment, serving as the baseline assessment. Cases received two additional mailings with returns, 6 and 12 months after the baseline. Non-responders were reminded after two weeks to increase the response rate. First-patient-in was in March 2022 and last-patient-out was in November 2023.

Recruitment and selection of study subjects

Patients were consecutively recruited between March and October 2022 in 14 general medical practices in Munich and the greater Munich area. Medical assistants or the participating medical student asked patients in the waiting rooms if they wanted to participate in the study and checked the inclusion criteria of age \geq 18 years and sufficient German language skills. Eligible patients were fully informed about the study and their rights orally and in writing by their general practitioners and included upon giving written consent.

Participation of general practitioners and of a citizen advisory board is described in supplementary data S2.

Measurements and outcomes

The questionnaires enquired about socio-demographics: age (years), height (cm), weight (kg) and gender (female/male/diverse). Standardized and validated questionnaires assessed anxiety, depression, and somatic symptom disorder (SSD), alongside symptom surveys for fatigue and dyspnoea to gauge symptom presence and severity, or somatic and psychosomatic comorbidity: Patient Health Questionnaire-15 (PHQ-15, [16]), Patient Health Questionnaire-4 (PHQ-4, [17]), Somatic Symptom Disorder - B Criteria Scale (SSD-12, [18]), Fatigue Assessment Scale (FAS, [19]), and modified dyspnoea scale of the Medical Research Council (mMRC, [20, 21]) (see supplementary data S3 and the original publications of these questionnaires for details).

Presence of well-described and common symptoms was also queried (each yes/no) [3, 22]. These were fatigue, shortness of breath, chest pain, memory problems, concentration problems, sleep disturbance, palpitations, dizziness, depression, anxiety, tinnitus, loss of appetite, weight loss, cough, headache, change in sense of smell, change in sense of taste, skin rashes and muscle pain. Cases were additionally asked about the timing of their SARS-CoV-2 infection (month and year) and other factors that were considered to be related to the symptom burden of the COVID-19 disease or to result from it, that is vaccination status at the time of infection (yes/no), additional infections between surveys (yes/no), and daily life impairment (yes/no). No repeated symptom surveys were carried out for the controls, as it was likely that they would also become infected in between the survey times. More importantly, without reference to an infection date in the control group, we would not have been

able to obtain any meaningful information about a timedependent course. Vaccination status was also not investigated in the controls, as the research question relating to vaccination was its association with the specific burden of the COVID-19 disease, which was not present in the controls.

Data management is described in supplementary data S4.

Statistical analysis

The distribution of data is presented by descriptive statistics. Trends over time are described for six-month periods since month of infection in accordance with the intervals between the survey times. The cohort of cases was further divided into sub-cohorts most likely infected with the Omicron variant or earlier variants, as these showed markedly different effects. The division was based on 31 December 2021 on which the Omicron variant almost completely and immediately replaced the earlier variants in Germany [23]. Group differences were tested using t-tests, analysis of variance (ANOVA) and chi-squared tests.

Multivariable binary logistic regression models were used to estimate the time-dependent frequency of reported symptoms in relation to the SARS-CoV-2 infection status (yes 'Omicron'/yes 'earlier variant'/no), the time measured in months since infection (corresponding to a value of zero in the controls) and their interaction. Other predictors of interest were the socio-demographics and the reported psychosomatic comorbidity. The inclusion of height and weight was preferred to the inclusion of body mass index (BMI) in order to improve the goodness of fit of the models [24]. The models were also extended to analyse the relation of the burden of the COVID-19 disease to vaccination prior to infection and further infections that occurred between the survey times in the cohorts of cases. All models were pre-specified. The occurrence of any symptoms or the frequent symptoms of fatigue, dyspnoea, impaired concentration and memory problems were considered to be of primary interest. Effect estimates are expressed as odds ratios (OR) or translated into marginal event probabilities. Respective hypothesis tests and 95% confidence intervals were adjusted for repeated measures using a clustered covariance matrix for the models' parameter estimates [25, 26].

Descriptive network analyses were constructed to illustrate bivariate correlations and partial correlations [27]. To increase clarity and interpretability, only correlations of moderate size ($|\mathbf{r}| \ge 0.3$) are displayed.

Statistical analysis was performed using R 4.3.0 (The R Foundation for Statistical Computing, Vienna, Austria). All tests were two sided with exploratory 5% significance levels. Sample size considerations are given in supplementary data S5.

Results

Study population

Included were n=204 cases (n=141 Omicron; n=63 earlier variants), and n=119 controls. Cases reported infections between March 2020 and August 2022 (Fig. 1A). The reported infection times aligned with the incidence of COVID-19 in Germany during this period [28]. Recruitment and follow-up resulted in survey times reaching from 0 to 21 months and 3 to 42 months since infection, respectively (Fig. 1B). There were 145 (71.1%) and 146 (71.6%) cases responding at the two follow-up survey times and 167 (81.9%) responding at least once.

The control cohort was more than 10 years older (p < 0.001), the Omicron cohort included about 8–9% more women (p=0.308) and distribution of weight and height were comparable (Table 1). Among the cases, the Omicron cohort more frequently reported a previous vaccination (p < 0.001), which may be due to the increasing availability of the vaccine over the course of the pandemic. No diverse gender was reported; thus, the variable



Fig. 1 (A) Frequency distribution of reported infections between March 2020 and August 2022. Colour indicates the division of cases into a cohort with Omicron infection and a cohort with infection by earlier variants using 31 December 2021 as the cut-off date. (B) Frequency distribution of the survey times as months since the reported infection, covering a total of 42 months

Table 1	Socio-d	lemograp	hic da	ta of i	the co	horts at	base	line.
Indices c	give the I	number o	f missi	ng va	alues			

	Controls	Omicron	Earlier vari-	<i>p</i> -
	(<i>n</i> = 119)	(<i>n</i> = 141)	ants (<i>n</i> = 63)	value
Age (years)	59.1 ± 14.9^{1}	48.6±15.1	48.2 ± 16.7^{1}	< 0.001
Sex (female)	71 (59.7%)	95 (67.4%)	36 (58.1%) ¹	0.308
Height	171±9	170±9	171±9	0.706
Weight	74.5 ± 19.4^{3}	76.4 ± 21.1^{1}	77.6 ± 18.1^2	0.581
Previous vaccination	-	135 (96.4%) ¹	24 (38.7%) ¹	< 0.001

was converted to sex (female/male). All variables presented here were included in the multivariable analysis.

Descriptive statistics

The frequency of any symptoms or the symptoms of fatigue, dyspnoea, impaired concentration and memory problems, and DLI, are shown in Table 2. Direct cohort comparison was feasible for up to two years after infection. Results show that disease symptoms were substantially more frequent in cases compared to controls. Over time, this frequency decreased in the Omicron cohort,

but remained high in the cohort with infections with earlier virus variants. High frequency and differing trends could also be observed for DLI. Statistics for all investigated symptoms and the entire observation period are available in supplementary data S6. These results reflect the effects described above and show the persistence of symptoms in the earlier virus variant cohort over a long period (supplementary data S6).

Cases also exhibited higher frequency of mMRC (moderate/severe) and higher sum scores of SSD-12, FAS, PHQ-15 and PHQ-4, assessing perceived dyspnoea, fatigue, somatisation disorders, anxiety and depression (Table 2). No obvious time trends could be recognised in this analysis. Statistics on the whole observation period are available in supplementary data S6.

Multivariable regression models

Probabilities of the occurrence of any symptoms, fatigue, impaired concentration and memory problems, estimated by multivariable regression models, are shown in Fig. 2. The models are presented in Table 3. Despite being frequently observed, dyspnoea could not be considered

Table 2 Descriptive statistics on disease symptoms defined to be of primary interest, DLI, psychosomatic comorbidity, mMRC and FAS, per cohort and summarised for four half-years since infection, in which a direct comparison is possible. Statistics for further symptoms and up to the eighth half-year are presented in supplementary data S6. Indices give the number of missing values

			Time since infection in half-years (N of Omicron/Earlier variants)				
	Controls (N = 119)	Cases	1	2	3	4	
			(124/25)	(108/27)	(113/33)	(12/23)	
Disease symptoms							
Any symptom	50 (43%)	Omicron	90 (74%)	59 (55%)	58 (52%) ¹	5 (42%)	
		Earlier	19 (79%)	21 (78%)	20 (61%)	15 (68%)	
Fatigue	19 (16%)	Omicron	59 (48%) ¹	43 (40%)	41 (37%) ²	3 (25%)	
		Earlier	12 (48%)	13 (50%) ¹	11 (33%)	13 (57%)	
Concentration/memory*	14 (12%) ¹	Omicron	38 (31%) ³	26 (24%)	22 (20%) ¹	3 (25%)	
		Earlier	11 (48%) ²	11 (41%)	11 (33%)	9 (41%)	
Impaired concentration	11 (9%) ¹	Omicron	38 (31%) ³	22 (20%)	21 (19%) ¹	3 (25%)	
		Earlier	10 (43%) ²	10 (37%)	10 (30%)	9 (39%)	
Memory problems	9 (8%) ¹	Omicron	22 (18%) ³	18 (17%)	17 (15%) ¹	2 (17%)	
		Earlier	10 (43%) ²	8 (30%)	10 (30%)	6 (29%) ²	
Dyspnoea	12 (10%) ¹	Omicron	31 (26%) ⁵	13 (12%)	13 (12%) ²	1 (8%)	
		Earlier	9 (36%)	5 (19%)	9 (27%)	7 (32%) ¹	
DLI	NA	Omicron	51 (41%)	32 (30%) ³	28 (25%) ¹	3 (25%)	
		Earlier	11 (44%)	7 (26%)	12 (39%) ²	11 (48%)	
Psychosomatic comorbidity							
SSD-12	7.8 ± 8.3^{7}	Omicron	11.2 ± 10.1	11.5 ± 10.9^2	11.2 ± 10.8	15.2 ± 11.4	
		Earlier	13.5 ± 10.2	10.9 ± 9.9	12.7 ± 10.7^{1}	13.9 ± 9.6	
PHQ-15	5.2 ± 4.4^{7}	Omicron	7.0 ± 5.2^{3}	6.1 ± 5.2^{11}	6.0 ± 5.2^4	6.0 ± 4.6^2	
		Earlier	7.3 ± 6.7^{1}	6.9 ± 4.9^{1}	6.5 ± 5.4^{3}	9.0 ± 7.5^{1}	
PHQ-4	1.8 ± 2.1^{1}	Omicron	2.6 ± 2.7	2.5 ± 2.9^2	2.2 ± 2.6^{1}	2.6 ± 2.5	
		Earlier	2.8 ± 2.5	2.1 ± 2.6	2.4 ± 2.5	3.2 ± 2.6^{1}	
mMRC (moderate/severe)	4 (4%) ⁷	Omicron	18 (16%) ¹²	12 (13%) ¹⁶	8 (8%) ¹⁵	1 (10%) ²	
		Earlier	2 (8%)	4 (17%) ³	3 (11%) ⁶	4 (20%) ³	
FAS	18.2 ± 5.7^{12}	Omicron	22.9±8.2	21.4 ± 7.8^{1}	21.3 ± 7.4	22.4 ± 7.2	
		Earlier	24.6 ± 10.6	23.3±8.2	23.6 ± 8.3^{1}	24.0 ± 8.7^{1}	





Fig. 2 Probability (solid line) and pointwise 95% confidence intervals (background colouring) of the outcomes 'any symptom' (**A**), fatigue (**B**) and impaired concentration or memory problems (**C**) for each cohort in dependence of time measured as months since infection. Ticks indicate the frequency distribution of outcomes (top = 'outcome reported', bottom = 'outcome not reported'). Estimates are derived from the multivariable logistic regression models reported in Table 3

in this analysis as the number of records was not sufficient to fit the large multivariable model. Case cohorts with Omicron or earlier virus variant infection were more likely to report any disease symptom in the initial months after infection compared to controls, with estimated OR of 4.15 (p<0.001) and 3.51 (p=0.054) in the month of infection. A decreasing trend was observed in the Omicron cohort (OR 0.89 per month, p < 0.001), which approached the control cohort over time, while the earlier virus variant cohort showed a persistent effect (OR 1.00 per month, p=0.903). Psychosomatic comorbidity was associated with an increased risk, with the PHQ-15 reaching statistical significance (OR 1.21, p < 0.001). Additional analyses of case cohorts regarding prior vaccination or further infections between the survey times showed numerical increases in risk (OR 1.18, *p*=0.800 and OR 2.63, *p*=0.035; models not shown).

Network analysis

Results of the network analysis in case cohorts revealed numerous moderate to strong correlations of DLI with disease symptoms and psychosomatic comorbidity (Fig. 3A; see supplementary data S7 for all numerical values). Strongest correlations were observed with fatigue (r=0.628), SSD-12 (r=0.583), dyspnoea (r=0.539) and

PHQ-15 (r=0.515). The only significant direct or partial correlation, meaning that it could not be explained by indirect correlations via other variables, existed with fatigue (r=0.338, Fig. 3B; see also supplementary data S7 for all numerical values).

Discussion

Main findings

We found symptom persistence in more than half of the patients after SARS-CoV-2 infection, recruited from primary care. As a concomitant effect, these patients reported considerably more physical comorbidity than controls, independently of an equally substantially higher level of psychological comorbidity. DLI was identified as another prevalent problem showing strongest correlations with fatigue, dyspnoea, SSD-12 and PHQ-15. These effects were particularly persistent with earlier virus variants.

Strengths and limitations

Inclusion of controls enables symptom frequency comparison. To our knowledge, there is no comparative evaluation in primary care up to now. However, a nocebo effect cannot be ruled out since a 'placebo infection' cannot be simulated in the controls. Speculation could arise **Table 3** Multivariable binary logistic regression models including a presentation of odds ratios with 95% confidence intervals (95%-CI). *, Concentration/memory" =, Impaired concentration' or, Memory problems'. ^aTime since infection with Omicron variant. ^bTime since infection with earlier variants

		Regression	Regression coefficients			Odds ratios		
Outcome	Predictor variables	Estimate	Std. error	p-value	Estimate	95%-Cl Lower	95%-Cl Upper	
Any Symptom	Intercept	2.568	3.760	0.495				
	Earlier variants	1.256	0.653	0.054	3.512	0.537	22.948	
	Omicron	1.424	0.360	< 0.001	4.154	1.474	11.704	
	PHQ-15	0.193	0.045	< 0.001	1.213	1.065	1.381	
	SSD-12	0.037	0.019	0.060	1.037	0.981	1.097	
	FAS	0.045	0.027	0.101	1.046	0.967	1.132	
	mMRC (moderate/severe)	-0.118	0.638	0.854	0.889	0.142	5.569	
	PHQ-4	0.058	0.079	0.465	1.060	0.844	1.331	
	Age (years)	0.003	0.009	0.712	1.003	0.977	1.031	
	Height (cm)	-0.031	0.022	0.160	0.970	0.910	1.033	
	Sex (male)	0.181	0.419	0.666	1.198	0.360	3.993	
	Weight (kg)	-0.003	0.009	0.748	0.997	0.973	1.022	
	Time (month) x Omicron ^a	-0.112	0.021	< 0.001	0.894	0.841	0.950	
	Time (month) x Earlier ^b	-0.004	0.035	0.903	0.996	0.900	1.102	
Fatigue	Intercept	0.749	3.747	0.842				
	Earlier variants	1.125	0.698	0.107	3.079	0.415	22.840	
	Omicron	1.521	0.452	0.001	4.578	1.252	16.734	
	PHQ-15	0.111	0.036	0.002	1.117	1.007	1.239	
	SSD-12	0.023	0.021	0.278	1.023	0.963	1.087	
	FAS	0.085	0.022	< 0.001	1.088	1.020	1.161	
	mMRC (moderate/severe)	0.583	0.467	0.212	1.792	0.469	6.841	
	PHQ-4	0.042	0.073	0.570	1.042	0.845	1.286	
	Age (years)	0.003	0.010	0.804	1.003	0.973	1.033	
	Height (cm)	-0.030	0.021	0.165	0.971	0.913	1.032	
	Sex (male)	-0.385	0.436	0.378	0.680	0.194	2.381	
	Weight (kg)	-0.005	0.007	0.529	0.995	0.974	1.017	
	Time (month) x Omicron ^a	-0.058	0.026	0.026	0.944	0.876	1.017	
	Time (month) x Earlier ^b	0.010	0.039	0.796	1.010	0.904	1.129	
Concentration/memory*	Intercept	-4.908	4.022	0.222				
	Earlier variants	1.816	0.618	0.003	6.145	1.043	36.216	
	Omicron	0.923	0.484	0.057	2.516	0.626	10.106	
	PHQ-15	0.078	0.039	0.044	1.081	0.967	1.207	
	SSD-12	0.048	0.020	0.017	1.049	0.990	1.112	
	FAS	0.108	0.025	< 0.001	1.115	1.037	1.198	
	mMRC (moderate/severe)	-0.503	0.387	0.194	0.605	0.199	1.839	
	PHQ-4	-0.053	0.064	0.404	0.948	0.789	1.139	
	Age (years)	0.009	0.010	0.382	1.009	0.980	1.040	
	Height (cm)	0.001	0.023	0.971	1.001	0.937	1.069	
	Sex (male)	-0.315	0.493	0.524	0.730	0.177	3.010	
	Weight (kg)	-0.011	0.009	0.221	0.990	0.965	1.014	
	Time (month) x Omicron ^a	-0.050	0.028	0.077	0.951	0.878	1.031	
	Time (month) x Earlier ^b	-0.046	0.031	0.142	0.955	0.873	1.045	

that symptoms may have developed solely due to knowledge of infection. However, this seems unlikely given the large differences between cohorts, symptom persistence and the typical symptom patterns of fatigue, dyspnoea and impaired concentration. Furthermore, a selection bias cannot be ruled out in this observational study, as the controls were on average about 10 years older than the cases. However, it is known that older age is a risk factor for the disease [1, 14]. Therefore, the even better results in the older controls suggest that the derived differences between the groups may even be conservative estimates. In addition, the group comparisons in



Fig. 3 (A) Bivariate and (B) partial correlations with a size of at least $|r|\ge 0.3$ between DLI (yellow), disease symptoms (orange), mMRC, FAS and psychosomatic comorbidity (blue) and prior vaccination (green). Increasing size of correlation is indicated by closer neighbour-ship and width of edges. Numeric values of correlations are given in supplementary data S7

the multivariable analyses were adjusted for age and other sociodemographic data to reduce potential bias. The questionnaire-based survey may have led to recall bias, especially regarding infections and vaccination, and information bias, for example due to question misunderstanding. These sources of bias may have led to both an underestimation and an overestimation of effects, for example in relation to the potentially protective or harmful effects of vaccination and several previous infections, or in relation to the frequency of symptoms. Even with 81.9% of the cases responding in the follow-up, attrition bias remains possible, which can potentially lead to an accumulation or reduction of more severely affected persons and consequently to over- or underestimation of symptom frequency. However, we did not observe a decrease in responses between the second and third survey, with many participants responding to one or the other survey. The division of cases into sub-cohorts may have led to allocation bias and an underestimation of respective group differences. However, the Omicron variant almost completely and immediately displaced earlier variants in Germany around 31 December 2021 [23], and only 9 (4%) and 18 (9%) cases reported an infection in December 2021 and January 2022, respectively. The complex multivariable analyses could not be carried out for each symptom due to the study's limited sample size. Corresponding models were intentionally not created to prevent bias. The severity of disease could not be taken into account, for example by use of the National Institutes of Health (NIH) clinical spectrum of SARS-CoV-2 infection ranging from no symptoms to critical illness, as clinical assessment or imaging was not available in the present study [29]. The control questionnaire excluded COVID-19-specific questions about vaccination status and DLI. In retrospect, this information could have been used for additional analyses that go beyond the research question of the relation to the burden of the COVID-19 disease. No repeated surveys were conducted on controls, as the trend over time was linked to the infection event. Possible effects due to the calendar timeline could therefore not be controlled.

Relation to existing literature

Population-based surveys indicate DLI prevalence of 5–28% in patients 12 to 18 months after SARS-CoV-2 infection [30]. DLI frequency is significantly higher in our primary care study, exceeding 50%. This aligns with healthcare system selection mechanisms [31], as more patients naturally consult their general practitioner for persistent comorbidity. Patient suffering is evident when compared to controls. Cases had considerably more symptoms, greater impairment of everyday life and markedly higher psychological comorbidity, although the controls were on average 10 years older. This clear difference is remarkable given that there are studies in which the so-called long COVID or post-COVID-19 syndrome

symptoms are attributed to placebo or, better, nocebo effects [32–34].

The comparison with controls could contribute to better understanding of LCS or PCS. Of course, psychological effects play a major role in the development of symptoms after infections. On the one hand, increased psychological comorbidity is a predictor for the development of post-infectious symptoms, as has also been shown by routine data analysis for other infectious diseases [35, 36]. However, these analyses indicate psychosomatic comorbidity as an independent predictor, showing no significant interaction with the infectious event. In addition, expectations play a significant role, as negative symptom expectations and processing can create tension that almost invites symptoms to develop [32], which is accompanied by excessive preoccupation with the disease in terms of somatic symptom disorder. However, a pure nocebo effect seems unlikely, as the controls had fewer symptoms and a significantly lower disease burden. Rather, it seems obvious that symptoms, especially fatigue, significantly contribute to the patients' impaired mental state as indicated by our network analysis and results of our earlier population-based study [11].

Implications for research and practice

Notably, earlier variants exhibit persistent fatigue, dyspnoea, and concentration impairment, whereas the Omicron variant shows quicker symptom decline. This could contribute to a more differentiated assessment and prognosis of the time course of individual patients, although further research on symptom development over time and risk factors is required. These findingsmay be attributed to vaccination effects [37], with broader population coverage during the Omicron variant emergence, or increasing infestation [38]. However, the extent of mental comorbidity remains rather constant over time. As vaccination and infestation continue to offer partial protection at best, managing PCS remains a challenge in primary care, particularly due to the lack of specific treatments. Impaired mental health might at least partly be explained by persistent unexplained and subjectively disturbing symptoms in the new clinical picture of PCS, which are challenging to treat [11]. Therefore, in addition to treating the symptoms, the high level of psychological distress should also be considered, in line with the bio-psychosocial model, corresponding to general practitioner heuristics [39]. Numerous studies, some of which could also be suitable for GP practices, are currently being conducted to find out which psychological interventions can best support patients [40]. Beyond that, development of patient guidelines and manuals could also be useful to enable patient empowerment [41, 42]. Finally, as far as fatigue is concerned, staged activation and pacing appear to be the most promising [3]. Further studies would need to investigate how this can best be achieved in practice, for example using activating training apps [43]. In addition to stepwise activation and pacing [3], appropriate psychological interventions could be useful in this regard.

Conclusion

The multitude of complex relations and independent effects of disease symptoms, psychosomatic comorbidity and DLI emphasizes that patients of general practitioners need holistic treatment. The symptom of fatigue may have a key role in establishing suitability for everyday life. Therefore, stepwise activation and pacing [3] and additional appropriate psychological interventions may be indicated in these difficult-to-treat patients.

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s12875-024-02551-w.

Supplementary Material 1

Acknowledgements

We thank the general practices, their staff and patients for their support and participation in the study. Some of the participating practices are also members of the Bavarian Practice Based Research Network (BayFoNet, www. bayfonet.de). We acknowledge all members of the patient and citizen advisory board of the Institute of General Practice and Health Services Research of the Technical University of Munich, coordinated by Clara Teusen and Stefanie Eck, for their feedback on the study results and their presentation.

Author contributions

A.S. developed the study concept, designed the study, interpreted the study results and contributed significantly to the writing of the manuscript. A.H. contributed to the design of the study, performed the statistical analysis, interpreted the results and drafted the manuscript. J.D. performed the data collection and cleaning, participated in the data analysis and contributed substantially to the writing of the manuscript. C.T. and S.E. contributed to the interpretation of the results and the writing of the manuscript. All authors read and approved the final manuscript.

Funding

The authors declare no relevant funding. Open Access funding enabled and organized by Projekt DEAL.

Data availability

The written consent obtained from patients does not authorise the disclosure of study data to third parties, for which there is no legal basis. However, the extent to which anonymised data, e.g. aggregated data, may be shared can be reviewed upon reasoned request to the corresponding author.

Declarations

Ethics approval and consent to participate

The study was approved by the local ethics committee of the Technical University of Munich before conduct (Reference number: 722/21 S-SR). Eligible patients were fully informed about the study and their rights orally and in writing by their general practitioners. All participants in the study gave written informed consent. The study was conducted in accordance with the Declaration of Helsinki.

Consent for publication

Not applicable.

Competing interests

The authors declare no competing interests.

Received: 17 May 2024 / Accepted: 30 July 2024 Published online: 10 August 2024

References

- 1. Lippi G, Sanchis–Gomar F, Henry BM. COVID-19 and its long-term sequelae: what do we know in 2023? Polish Archives of Internal Medicine; 2023.
- Contreras S, Iftekhar EN, Priesemann V. From emergency response to longterm management: the many faces of the endemic state of COVID-19. Lancet Reg Health Eur. 2023;30:100664.
- Koczulla AR, Ankermann T, Behrends U, et al. [German S1 Guideline Long-/ Post-COVID]. Pneumologie. 2022;76(12):855–907.
- NICE. COVID-19 rapid guideline: managing the longterm effects of COVID-19. National Institute for Health and Care Excellence (NICE), Scottish Intercollegiate Guidelines Network (SIGN) and Royal College of General Practitioners (RCGP) 2022.
- Lund LC, Hallas J, Nielsen H, et al. Post-acute effects of SARS-CoV-2 infection in individuals not requiring hospital admission: a Danish population-based cohort study. Lancet Infect Dis. 2021;21(10):1373–82.
- Radtke T, Ulyte A, Puhan MA, et al. Long-term symptoms after SARS-CoV-2 infection in children and adolescents. JAMA. 2021;326(9):869–71.
- Wulf Hanson S, Abbafati C, Aerts JG, et al. Estimated global proportions of individuals with persistent fatigue, cognitive, and respiratory symptom clusters following symptomatic COVID-19 in 2020 and 2021. JAMA. 2022;328(16):1604–15.
- Tran VT, Porcher R, Pane I, et al. Course of post COVID-19 disease symptoms over time in the ComPaRe long COVID prospective e-cohort. Nat Commun. 2022;13(1):1812.
- Joffe AR, Elliott A. Long COVID as a functional somatic symptom disorder caused by abnormally precise prior expectations during bayesian perceptual processing: a new hypothesis and implications for pandemic response. SAGE Open Med. 2023;11:20503121231194400.
- Blankenburg J, Wekenborg MK, Reichert J, et al. Comparison of mental health outcomes in seropositive and seronegative adolescents during the COVID19 pandemic. Sci Rep. 2022;12(1):2246.
- Schneider A, Huber L, Lohse J, et al. Association between somatic symptom disorder and symptoms with daily life impairment after SARS-CoV-2 infection - results from a population-based cross-sectional study. J Psychosom Res. 2023;168:111230.
- 12. Greißel A, Schneider A, Donnachie E, et al. Impact of pre-existing mental health diagnoses on development of post-COVID and related symptoms: a claims data-based cohort study. Sci Rep. 2024;14(1):2408. 2024/01/29.
- Gao P, Liu J, Liu M. Effect of COVID-19 vaccines on reducing the risk of long COVID in the Real World: a systematic review and Meta-analysis. Int J Environ Res Public Health. 2022;19(19).
- 14. Fumagalli C, Zocchi C, Tassetti L, et al. Factors associated with persistence of symptoms 1 year after COVID-19: a longitudinal, prospective phone-based interview follow-up cohort study. Eur J Intern Med. 2022;97:36–41.
- 15. Miyazato Y, Morioka S, Tsuzuki S, et al. Prolonged and late-onset symptoms of Coronavirus Disease 2019. Open Forum Infect Dis. 2020;7(11):ofaa507.
- Kroenke K, Spitzer RL, Williams JB. The PHQ-15: validity of a new measure for evaluating the severity of somatic symptoms. Psychosom Med. 2002 Mar-Apr;64(2):258–66.
- Löwe B, Wahl I, Rose M, et al. A 4-item measure of depression and anxiety: validation and standardization of the Patient Health Questionnaire-4 (PHQ-4) in the general population. J Affect Disord. 2010;122(1–2):86–95.
- Toussaint A, Murray AM, Voigt K, et al. Development and validation of the somatic Symptom Disorder-B criteria scale (SSD-12). Psychosom Med. 2016;78(1):5–12.
- 19. De Vries J, Michielsen H, Van Heck GL, et al. Measuring fatigue in sarcoidosis: the fatigue Assessment Scale (FAS). Br J Health Psychol. 2004;9(Pt 3):279–91.
- Fletcher CM. The clinical diagnosis of pulmonary emphysema; an experimental study. Proc R Soc Med. 1952;45(9):577–84.
- Bestall JC, Paul EA, Garrod R, et al. Usefulness of the Medical Research Council (MRC) dyspnoea scale as a measure of disability in patients with chronic obstructive pulmonary disease. Thorax. 1999;54(7):581–6.

- Soriano JB, Murthy S, Marshall JC, et al. A clinical case definition of post-COVID-19 condition by a Delphi consensus. Lancet Infect Dis. 2022;22(4):e102–7.
- RKI. Weekly situation report of the RKI on the coronavirus disease 2019. Wöchentlicher Lagebericht Des RKI Zur Coronavirus-Krankheit-2019 (COVID-19) 03.02.2022 –. AKTUALISIERTER STAND FÜR DEUTSCHLAND; 2022.
- Shuey MM, Huang S, Levinson RT, et al. Exploration of an alternative to body mass index to characterize the relationship between height and weight for prediction of metabolic phenotypes and cardiovascular outcomes. Obes Sci Pract. 2022;8(1):124–30.
- 25. Zeileis A. Object-oriented computation of Sandwich estimators. J Stat Softw. 2006;08(15):1–16.
- 26. Zeileis A, Köll S, Graham N. Various versatile variances: an object-oriented implementation of clustered covariances in R. J Stat Softw. 2020;10(07):1–36.
- Schafer JO-RR, Zuber V, Ahdesmaki M, Silva APD. Strimmer. K. corpcor: Efficient Estimation of Covariance and (Partial) Correlation_. R package version 1.6.10, 2021.
- RKI. Weekly situation report of the RKI on the coronavirus disease 2019. Wöchentlicher Lagebericht Des RKI Zur Coronavirus-Krankheit-2019 (COVID-19) 08.06.2023 –. AKTUALISIERTER STAND FÜR DEUTSCHLAND; 2023.
- COVID-19 Treatment Guidelines Panel. Coronavirus Disease 2019 (COVID-19) Treatment Guidelines. National Institutes of Health. [Internet]. 2024 [cited 22.07.2024]. https://www.covid19treatmentguidelines.nih.gov/
- 30. Peter RS, Nieters A, Kräusslich HG, et al. Post-acute sequelae of covid-19 six to 12 months after infection: population based study. BMJ. 2022;379:e071050.
- Green LA, Fryer GE Jr., Yawn BP, et al. The ecology of medical care revisited. N Engl J Med. 2001;344(26):2021–5.
- Lemogne C, Gouraud C, Pitron V, et al. Why the hypothesis of psychological mechanisms in long COVID is worth considering. J Psychosom Res. 2023;165:111135.
- Matta J, Robineau O, Wiernik E et al. Depression and anxiety before and at the beginning of the COVID-19 pandemic and incident persistent symptoms: a prospective population-based cohort study. Mol Psychiatry. 2023 Jul 19.
- Selvakumar J, Havdal LB, Drevvatne M, et al. Prevalence and characteristics Associated with Post-COVID-19 Condition among Nonhospitalized adolescents and Young adults. JAMA Netw Open. 2023;6(3):e235763.
- 35. Donnachie E, Schneider A, Enck P. Comorbidities of patients with functional somatic syndromes before, during and after first diagnosis: a Population-based study using bavarian Routine Data. Sci Rep. 2020;10(1):9810.
- Donnachie E, Schneider A, Mehring M, et al. Incidence of irritable bowel syndrome and chronic fatigue following GI infection: a population-level study using routinely collected claims data. Gut. 2018;67(6):1078–86.
- Ceban F, Kulzhabayeva D, Rodrigues NB, et al. COVID-19 vaccination for the prevention and treatment of long COVID: a systematic review and metaanalysis. Brain Behav Immun. 2023;111:211–29.
- RKI. SARS-CoV-2. Basic virological data and virus variants in the period 2020–2022. SARS-CoV-2: Virologische Basisdaten Sowie Virusvarianten Im Zeitraum Von 2020–2022. 2023.
- Wonca. The european definition of gGeneral practice / family medicine. wonca europe 2023 edition. 2023.
- Hawke LD, Nguyen ATP, Ski CF, et al. Interventions for mental health, cognition, and psychological wellbeing in long COVID: a systematic review of registered trials. Psychol Med. 2022;52(13):2426–40.
- 41. Brown K, Yahyouche A, Haroon S, et al. Long COVID and self-management. Lancet. 2022;399(10322):355.
- 42. Greenhalgh T, Sivan M, Delaney B, et al. Long covid-an update for primary care. BMJ. 2022;378:e072117.
- Samper-Pardo M, León-Herrera S, Oliván-Blázquez B, et al. Effectiveness of a telerehabilitation intervention using ReCOVery APP of long COVID patients: a randomized, 3-month follow-up clinical trial. Sci Rep. 2023;13(1):7943.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.